\documentclass[12pt]{article}
\usepackage{amsmath,,calrsfs}
\usepackage{amsfonts}
\usepackage{amssymb}
\usepackage{amscd}
\usepackage{bbm}
\usepackage{fancybox}
\usepackage{cite}
\usepackage{amsmath,amsfonts,amsbsy}
\usepackage{pstricks,pst-node}
\usepackage[small,bf,hang]{caption2}
%\usepackage{slashed}
\usepackage{graphicx}
\usepackage{epsfig}
\usepackage{psfrag}
\usepackage{comment}
%\usepackage{epstopdf}


\newcommand{\pf}[1]{\boldsymbol{\mathit{#1}}}

\usepackage{float}

\psset{unit=1.3cm,linewidth=.5pt,radius=.2}  % controls the size of diagrams



%\usepackage{enumitem}                       % fancier lists
\usepackage{multirow}                     % tables cells spanning multiple rows
\usepackage{float}                          % to force floaters to stay Here
\usepackage{lscape}                         % landscape laid out pages
%\usepackage[caption=false]{subfig}          % for subfigures
\usepackage{bm}


%%%%%%%%%%%%%%%%%%%
%  This produces documents of a size suitable for sending to a
% journal.
%\def\hybrid{\topmargin -20pt    \oddsidemargin 0pt
%        \headheight 0pt \headsep 0pt
%       \textwidth 6.5in        % US paper
%       \textheight 9in         % US paper
%        \textwidth 6.25in       % A4 paper
%       \textheight 9.5in       % A4 paper
%        \marginparwidth .875in
%        \parskip 5pt plus 1pt   \jot = 1.5ex}

\addtolength{\topmargin}{7mm} \textwidth  155mm \textheight 235mm
\setlength{\hoffset}{-1.1cm} \setlength{\voffset}{-2.6cm}
\renewcommand{\baselinestretch}{1.1}

%       The default is set to be hybrid
%\hybrid
%
%\newcommand{\pf}[1]{\boldsymbol{\mathit{#1}}}


\renewcommand{\topfraction}{0.99}
\renewcommand{\bottomfraction}{0.99}
\renewcommand{\textfraction}{0}
%

\newcommand{\mM}{{\mathcal M}}
\newcommand{\mN}{{\mathcal N}}

\newcommand{\cA}{{\cal A}}
\newcommand{\cB}{{\cal B}}
\newcommand{\cC}{{\cal C}}
\newcommand{\cD}{{\cal D}}
\newcommand{\cE}{{\cal E}}
\newcommand{\cF}{{\cal F}}
\newcommand{\cK}{{\cal K}}
\newcommand{\cL}{{\cal L}}
\newcommand{\cM}{{\cal M}}
\newcommand{\cN}{N}
\newcommand{\cO}{{\cal O}}
\newcommand{\cP}{{\cal P}}
\newcommand{\cQ}{{\cal Q}}
\newcommand{\cR}{{\cal R}}
\newcommand{\cS}{{\cal S}}
\newcommand{\cT}{{\cal T}}
\newcommand{\cV}{{\cal V}}
\newcommand{\cW}{{\cal W}}
\newcommand{\cZ}{{\cal Z}}
\newcommand{\ax}{\alpha}
\newcommand{\bx}{\beta}
\newcommand{\cx}{\gamma}
\newcommand{\dx}{\delta}
\newcommand{\ox}{\omega}
\newcommand{\ab}{\bar\alpha}
\newcommand{\bb}{\bar\beta}
\newcommand{\cb}{\bar\gamma}
\newcommand{\db}{\bar\delta}
\newcommand{\ib}{{\bar i}}
\newcommand{\jb}{{\bar j}}
\newcommand{\au}{\hat\alpha}
\newcommand{\bu}{\hat\beta}
\newcommand{\cu}{\hat\gamma}
\newcommand{\du}{\hat\delta}
\newcommand{\beq}{\begin{equation}}
\newcommand{\eeq}{\end{equation}}
\newcommand{\bi}{\begin{itemize}}
\newcommand{\ei}{\end{itemize}}
%\newcommand{\bea}{\begin{eqnarray}} %\newcommand{\eea}{\end{eqnarray}} %\newcommand{\ba}{\begin{array}} %\newcommand{\ea}{\end{array}}
\newcommand{\bt}{\begin{tabular}}
\newcommand{\et}{\end{tabular}}
\newcommand{\bc}{\begin{center}}
\newcommand{\ec}{\end{center}}
\newcommand{\half}{\frac12}
\newcommand{\quart}{\frac14}
\newcommand{\dslash}{\partial \hskip -.24truecm  / }
%\def\theequation{\arabic{section}.\arabic{equation}}
\def\Hnote#1{{\bf[HS: #1]}}
\def\Onote#1{{\bf[OH: #1]}}
\def\addref{{\bf [add ref]}}
\def\id{{\bf [improve discussion]}}
\def\ck{{\bf [check]}}
\def\xx{{\bf [xx]\ }}
\def\S{\hat S}
\def\N{\hat N}

\newcommand{\ket}[1]{|#1\rangle}
\newcommand{\bra}[1]{\langle#1|}
\newcommand{\bracket}[2]{\langle#1|#2\rangle}
\newcommand{\vev}[1]{\langle#1\rangle}
\newcommand{\re}{\text{Re}}
\newcommand{\im}{\text{Im}}
\newcommand{\ir}[1]{\hat#1}
\newcommand{\trans}[1]{{\vphantom{#1}}^t{#1}}
\newcommand{\ft}[2]{{\textstyle {\frac{#1}{#2}} }}
\newcommand{\bein}[3]{{#1}_{#2}^{\hspace{0.5em}{#3}}}

\DeclareMathOperator{\ad}{ad}
\newcommand{\g}{\mathfrak{g}}

\newcommand{\be}{\begin{equation}}
\newcommand{\ee}{\end{equation}}
\newcommand{\bea}{\begin{eqnarray}}
\newcommand{\eea}{\end{eqnarray}}
\newcommand{\ba}{\begin{array}}
\newcommand{\ea}{\end{array}}
\newcommand{\p}[1]{(\ref{#1})}
\newcommand{\lb}[1]{\label{#1}}
\newcommand{\vs}[1]{\vspace{#1 mm}}
\newcommand{\hs}[1]{\hspace{#1 mm}}
\def\bbox{{\,\lower0.9pt\vbox{\hrule \hbox{\vrule height 0.2 cm
\hskip 0.2 cm \vrule height 0.2 cm}\hrule}\,}}
\newcommand{\dsl}{\pa \kern-0.5em /}
\renewcommand{\t}{\theta}
\newcommand{\T}{\Theta}
\newcommand{\nn}{\nonumber \\}
\newcommand{\und}{\underline}


%%%%%%%%%%%%%%%%%%%% Macros Paul %%%%%%%%%%%%%%%%%%%%%%%%%%%%
%If you do not have the msbm fonts, delete the following 10 lines
\font\mybb=msbm10 at 10pt
\def\bb#1{\hbox{\mybb#1}}
\def\bZ {\bb{Z}}
\def\bR {\bb{R}}
\def\bE {\bb{E}}
\def\bT {\bb{T}}
\def\bM {\bb{M}}
\def\bL {\bb{L}}
\def\bG {\bb{G}}
\def\bH {\bb{H}}
\def\bC {\bb{C}}
\def\bA {\bb{A}}
\def\bX {\bb{X}}
\def\bP {\bb{P}}
\def\bJ {\bb{J}}
\def\bU {\bb{U}}
\def\bJ {\bb{J}}
\def\bV{\bb{V}}
\def\bW {\bb{W}}
\def\bI {\bb{I}}
\def\g{\gamma }
\def\t{\tau }
%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%%%%%%%
\newcommand{\bfa}{\mbox{\boldmath $\alpha$}}
\newcommand{\tbfa}{\mbox{\boldmath $\tilde \alpha$}}
\newcommand{\bfna}{\mbox{\boldmath $\nabla$}}
\def\bfo{\mbox{\boldmath $\omega$}}
\def\bfO{\mbox{\boldmath $\Omega$}}
\def\bfn{\mbox{\boldmath $\nabla$}}
\def\bfs{\mbox{\boldmath $\sigma$}}
\def\bfG{\mbox{\boldmath $\Gamma$}}
\def\bfth{\mbox{\boldmath $\theta$}}
\def\bfTh{\mbox{\boldmath $\Theta$}}
\def\bfPi{\mbox{\boldmath $\Pi$}}
\def\bfpi{\mbox{\boldmath $\pi$}}
\def\tr{{\rm tr}}
%%%%%%%%%%%%%%%%%


\def\eq#1{(\ref{#1})}

%%%%%%

%\makeatletter \@addtoreset{equation}{section} \makeatother
%\renewcommand{\theequation}{\thesection.\arabic{equation}}

\def\slashchar#1{\setbox0=\hbox{$#1$}           % set a box for #1
   \dimen0=\wd0                                 % and get its size
   \setbox1=\hbox{/} \dimen1=\wd1               % get size of /
   \ifdim\dimen0>\dimen1                        % #1 is bigger
      \rlap{\hbox to \dimen0{\hfil/\hfil}}      % so center / in box
      #1                                        % and print #1
   \else                                        % / is bigger
      \rlap{\hbox to \dimen1{\hfil$#1$\hfil}}   % so center #1
      /                                         % and print /
   \fi}

%%%%%

\def\eq#1{(\ref{#1})}
\newcommand{\w}[1]{\\[0.#1cm]}
\newcommand{\sh}{{\hat\sigma}}
\newcommand{\remark}[1]{{\bf [#1]}\marginpar[\hfill${\bf \Longrightarrow}$]
{${\bf \Longleftarrow}$} }


%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%%




\begin{document}
\begin{thebibliography}{10}







%\bibliographystyle{fullsort}
%\bibliography{etmgbib}



%\cite{Bergshoeff:2014pca}
\bibitem{Bergshoeff:2014pca}
  E.~Bergshoeff, O.~Hohm, W.~Merbis, A.~J.~Routh and P.~K.~Townsend,
  ``Minimal Massive 3D Gravity,''
  Class.\ Quant.\ Grav.\  {\bf 31}, 145008 (2014)
  [arXiv:1404.2867 [hep-th]].
  %%CITATION = ARXIV:1404.2867;%%


%\cite{Deser:1981wh}
\bibitem{Deser:1981wh}
S.~Deser, R.~Jackiw, and S.~Templeton, ``{Topologically Massive Gauge
  Theories},'' {\em Annals Phys.} {\bf 140} (1982)
372--411.
%%CITATION = APNYA,140,372;%%.

%\cite{Brown:1986nw}
\bibitem{Brown:1986nw}
  J.~D.~Brown and M.~Henneaux,
  ``Central Charges in the Canonical Realization of Asymptotic Symmetries: An Example from Three-Dimensional Gravity,''
{\em   Commun.\ Math.\ Phys}.\  {\bf 104} (1986) 207.
  %%CITATION = CMPHA,104,207;%%

%\cite{Li:2008dq}
\bibitem{Li:2008dq}
  W.~Li, W.~Song and A.~Strominger,
  ``Chiral Gravity in Three Dimensions,''
  JHEP {\bf 0804}, 082 (2008)
  [arXiv:0801.4566 [hep-th]].

    %\cite{Bergshoeff:2009hq}
\bibitem{Bergshoeff:2009hq}
  E.~A.~Bergshoeff, O.~Hohm and P.~K.~Townsend,
  ``Massive Gravity in Three Dimensions,''
  Phys.\ Rev.\ Lett.\  {\bf 102}, 201301 (2009)
  [arXiv:0901.1766 [hep-th]].
  %%CITATION = ARXIV:0901.1766;%%

\bibitem{Sinha}
  A.~Sinha,
  ``On the new massive gravity and AdS/CFT,''
  {\em JHEP} {\bf 1006} (2010) 061
  [arXiv:1003.0683].
  
 \bibitem{Paulos} 
  M.~F.~Paulos,
  ``New massive gravity extended with an arbitrary number of curvature corrections,''
  {\em Phys.\ Rev.}\ D {\bf 82} (2010) 084042
  [arXiv:1005.1646].
  
 

  %\cite{Arvanitakis:2014xna}
\bibitem{Arvanitakis:2014xna}
  A.~S.~Arvanitakis and P.~K.~Townsend,
  ``Minimal Massive 3D Gravity Unitarity Redux,''
  Class.\ Quant.\ Grav.\ {\bf 32}, 085003 (2015)  
  [arXiv:1411.1970 [hep-th]].
  %%CITATION = ARXIV:1411.1970;%%

  %\cite{Arvanitakis:2014yja}
\bibitem{Arvanitakis:2014yja} 
  A.~S.~Arvanitakis, A.~J.~Routh and P.~K.~Townsend,
  ``Matter Coupling in 3D `Minimal Massive Gravity',''
  Class.\ Quant.\ Grav.\  {\bf 31}, no. 23, 235012 (2014)
  [arXiv:1407.1264 [hep-th]].
%%CITATION = ARXIV:1407.1264;%%

%\cite{Arvanitakis:2015oga}
\bibitem{Arvanitakis:2015oga} 
  A.~S.~Arvanitakis, A.~Sevrin and P.~K.~Townsend,
  %``Yang-Mills as massive Chern-Simons theory: a third way to three-dimensional gauge theories,''
  Phys.\ Rev.\ Lett.\  {\bf 114}, no. 18, 181603 (2015)
  [arXiv:1501.07548 [hep-th]].
  %%CITATION = ARXIV:1501.07548;%%

  %\cite{Lovelock:1971yv}
\bibitem{Lovelock:1971yv}
  D.~Lovelock,
  ``The Einstein tensor and its generalizations,''
  J.\ Math.\ Phys.\  {\bf 12}, 498 (1971).
  %%CITATION = JMAPA,12,498;%%






  \end{thebibliography}


\begin{titlepage}%1
\begin{center}

\hfill  DAMTP-2015-31

\vskip 1cm




{\Large \bf The Third Way to 3D Gravity}




\vskip 1cm



{\bf Eric Bergshoeff\,${}^1$, Wout Merbis\,${}^2$,}
\vskip .2truecm

{\bf Alasdair J. Routh\,${}^3$ and Paul K. Townsend\,${}^3$} \\



\vskip 25pt




{\em $^1$ \hskip -.1truecm
\em Centre for Theoretical Physics,
University of Groningen, \\ Nijenborgh 4, 9747 AG Groningen, The Netherlands\vskip 5pt }

{email: {\tt e.a.bergshoeff@rug.nl}} \\

\vskip .4truecm

{\em $^2$ \hskip -.1truecm  Institute for Theoretical Physics, Vienna University of Technology, \\
Wiedner Hauptstrasse 8-10/136, A-1040 Vienna, Austria\vskip 5pt }

{email: {\tt merbis@hep.itp.tuwien.ac.at}} \\

\vskip .4truecm

{\em $^3$ \hskip -.1truecm
\em  Department of Applied Mathematics and Theoretical Physics,\\ Centre for Mathematical Sciences, University of Cambridge,\\
Wilberforce Road, Cambridge, CB3 0WA, U.K.\vskip 5pt }

{email: {\tt A.J.Routh@damtp.cam.ac.uk; p.k.townsend@damtp.cam.ac.uk}} \\





\end{center}

\vskip 0.5cm

\begin{center} {\bf ABSTRACT}\\[3ex]
\end{center}
{ \small
Consistency of Einstein's gravitational field equation $G_{\mu\nu} \propto  T_{\mu\nu}$ imposes a ``conservation condition'' on the $T$-tensor that is satisfied by (i) matter stress
tensors, as a consequence of the matter equations of motion,  and (ii) identically by certain other tensors,  such as the metric tensor. However, there is a third way, overlooked
until now because it  implies  a ``non-geometrical'' action: one {\it not} constructed  from the metric and its derivatives alone. The new possibility  is exemplified by the 3D ``minimal massive gravity'' model,  which resolves the ``bulk vs boundary'' unitarity problem of  topologically massive gravity  with anti-de Sitter asymptotics.  Although all known examples of the third way are in three spacetime dimensions, the idea is general and could, in principle, apply to higher-dimensional  theories.
}
\vskip 1cm

\begin{center}
Essay written for the Gravity Research Foundation \\ 2015 Awards for Essays on Gravitation.


\end{center}

\end{titlepage}
%%%%%%%%%%%%%%%%%%%%
%%%%%%%%%%%%%%%%%%%%%

\noindent
It is now 100 years since Einstein wrote down, after a long struggle,  his gravitational field equations, 
\begin{equation}\label{EFE}
G_{\mu\nu} = (8\pi G) \, T_{\mu\nu}\, ,
\end{equation}
where $T_{\mu\nu}$ is the matter stress tensor, $G$ is Newton's gravitational constant and $G_{\mu\nu}$ is  the Einstein tensor, defined in terms of the metric $g_{\mu\nu}$ and its Ricci tensor $R_{\mu\nu}$, and Ricci scalar $R$, by
\begin{equation}
G_{\mu\nu}= R_{\mu\nu} - \frac{1}{2} g_{\mu\nu} R\, .
\end{equation}
The  stress tensor satisfies, {\it as a consequence of the matter field equations}, the ``conservation condition''
\begin{equation}\label{conservation}
D^\mu T_{\mu\nu} =0\, ,
\end{equation}
where $D$ is the usual covariant derivative in a gravitational background. In  flat space this condition implies conservation of energy and momentum in the matter fields.
It was this condition that led Einstein to the Einstein tensor: he needed to construct from the metric a tensor $G_{\mu\nu}$  satisfying the Bianchi identity
\begin{equation}
D^\mu G_{\mu\nu} \equiv 0\, .
\end{equation}

However, one can turn the logic around: it is because the Einstein tensor satisfies the Bianchi identity
that the tensor $T_{\mu\nu}$ appearing on the right-hand side of the Einstein field equations (\ref{EFE}) must satisfy the conservation condition (\ref{conservation}).  This perspective leads to the following question. How many ways are there to construct  a tensor $T_{\mu\nu}$ satisfying
(\ref{conservation})?  Certainly, one may take $T_{\mu\nu}$ to be a matter stress tensor, but another obvious possibility  is to choose
$T_{\mu\nu} \propto  g_{\mu\nu}$.  The conservation condition is an {\it identity} for this tensor, which can be viewed as a dark-energy
contribution  to the  stress tensor, but it can also be taken over to the left-hand side to give us the modified field equation
\begin{equation}
G_{\mu\nu} + \Lambda g_{\mu\nu} = (8\pi G) T_{\mu\nu}\, ,
\end{equation}
where $\Lambda$ is a constant, the ``cosmological constant'' introduced by Einstein himself in 1917. But this is just a special case: for  any diffeomorphism
invariant functional $I[g]$,  the tensor
\begin{equation}
I_{\mu\nu} = \frac{1}{\sqrt{-\det g}} \frac{\delta I[g]}{\delta g^{\mu\nu}}
\end{equation}
satisfies the Bianchi-type identity
\begin{equation}
D^\mu I_{\mu\nu} \equiv 0\, .
\end{equation}
%Conversely, for any tensor $I_{\mu\nu}$ satisfying a Bianchi identity there is a corresponding action $I[g]$. 
Taking all such tensors over to the left hand side
we arrive at a generalisation of the Einstein field equations of the form
\begin{equation}
\label{eequalst}
E_{\mu\nu} = (8\pi G) T_{\mu\nu}\, ,
\end{equation}
where $E_{\mu\nu}$ is a symmetric tensor satisfying $D^\mu E_{\mu\nu}\equiv 0$. It will have the form
\begin{equation}
E_{\mu\nu} = \Lambda g_{\mu\nu} + \sigma G_{\mu\nu} + \dots \, ,
\end{equation}
where $\sigma$ is a dimensionless number and the omitted terms are higher-dimension tensors with coefficients that have  dimensions of increasing powers of  inverse mass. 

To return to our question, we have now seen that there are two standard ways to construct a tensor satisfying the conservation condition (\ref{conservation}). Either:
\begin{itemize}
\item[1.] The condition is satisfied {\it as a consequence of the matter field equations}, and the tensor contributes to the matter stress tensor,
\item[] or:
\item[2.] It is satisfied {\it identically} by means of a Bianchi-type identity, and the tensor contributes to the gravitational tensor $E_{\mu\nu}$.
\end{itemize}
However, there is one other logical possibility; a ``third way''.  Any abstract discussion would make it appear very unlikely that it could be realised in practice, so we will instead present an example. The example is  ``minimal massive gravity'' (MMG) \cite{Bergshoeff:2014pca},  which  describes an interacting massive graviton in three spacetime dimensions (3D).

Our starting point will be the third-order field equation of ``topologically massive gravity'' (TMG) \cite{Deser:1981wh},  which propagates a single spin-2 mode of mass $\mu$. Allowing for a cosmological constant, the TMG field equation is \eqref{eequalst} with
\begin{equation}\label{TMGE}
E_{\mu\nu} = \Lambda g_{\mu\nu} + \sigma G_{\mu\nu} + \frac{1}{\mu} C_{\mu\nu}\, .
\end{equation}
Here, $C_{\mu\nu}$ is the Cotton tensor, defined as
\begin{equation}
C_{\mu\nu} = \frac{1}{\sqrt{-\det g}} \, \varepsilon_\mu{}^{\tau\rho} D_\tau S_{\rho\nu}\, ,
\end{equation}
where the $S$-tensor is the 3D Schouten tensor
\begin{equation}
S_{\mu\nu} = R_{\mu\nu} - \frac{1}{4}g_{\mu\nu} R\, .
\end{equation}
The Cotton tensor is symmetric and traceless, and it satisfies a Bianchi identity; the corresponding action is the Lorentz-Chern-Simons action for the  affine connection.

Much of the interest in 3D gravity models derives from the potential simplifications of a lower dimension for the quantum theory. For 3D models with anti-de Sitter (AdS) asymptotics  there is the prospect of defining a quantum gravity theory via a holographically dual 2D conformal field theory (CFT).  A necessary condition for unitarity in this context is that the (left/right) central charges of the CFT (which differ in a parity-violating theory like TMG) are both positive. Additional unitarity requirements are that the bulk  spin-2 mode be non-tachyonic and have positive energy.  Given a 3D gravitational  action, all these requirements can be checked in a semi-classical approximation:  by linearization about an AdS background  for the bulk spin-2 mode,  and by consideration of the asymptotic symmetry algebra \cite{Brown:1986nw} for the CFT central charges.  It turns out that there is no choice of TMG parameters for which all these unitarity conditions are satisfied \cite{Li:2008dq} (and the same is true of 
``new massive gravity'', which propagates a parity doublet of massive spin-2 modes \cite{Bergshoeff:2009hq}).  This is known as the ``bulk-boundary clash'' because the bulk mode has negative energy when both central charges of the boundary CFT are positive,  although all bulk unitarity conditions must also have a CFT interpretation. Attempts to circumvent this problem by including higher-derivative modifications of the $E$-tensor (\ref{TMGE}) do not succeed \cite{Sinha}\cite{Paulos}.

However, there is a third way to modify the TMG equation. To see this, consider the tensor
\begin{equation}
J_{\mu\nu} = \frac{1}{2\det g}\ \varepsilon_\mu{}^{\rho\sigma} \varepsilon_\nu{}^{\tau\eta} S_{\rho\tau}S_{\sigma\eta}\, .
\end{equation}
This does not satisfy the conservation condition identically; in fact
\begin{equation}
\sqrt{-\det g}\, D_\mu J^{\mu\nu} = \varepsilon^{\nu\rho\sigma} S_\rho{}^\tau C_{\sigma\tau} \not\equiv 0\, .
\end{equation}
%However, if we use the source-free TMG equation in the form
%\begin{equation}\label{TMG2}
%C_{\mu\nu} = -(\sigma\mu) S_{\mu\nu} - \left(\mu\Lambda - \frac{1}{4} (\sigma\mu) R\right) g_{\mu\nu}\, ,
%\end{equation}
%then we find that
%\begin{equation}
%\varepsilon^{\nu\rho\sigma} S_\rho{}^\tau C_{\sigma\tau} = -(\sigma\mu)\,  \varepsilon^{\nu\rho\sigma} S^2_{\rho\sigma} -
%\left(\mu\Lambda - \frac{1}{4} (\sigma\mu) R\right) \varepsilon^{\nu\rho\sigma} S_{\rho\sigma} \equiv 0\, ,
%\end{equation}
However, if we use the source-free TMG equation to express the Cotton tensor as a linear combination of the Schouten tensor and the metric tensor, then we see that the right-hand-side of this equation vanishes identically. The $J$-tensor therefore satisfies the conservation condition {\it as a consequence of the TMG field equation}. This suggests the ``third way'':
\begin{itemize}
\item[3.] A tensor $T$ may satisfy  the conservation condition {\it as a consequence of the gravitational field equation itself}.
\end{itemize}
The obvious difficulty with this idea is that {\it we change the equation as soon as we include the new tensor as a source}. It seems that it would take a miracle for $J$ to continue to satisfy the conservation condition as a consequence of the gravitational field equation when $J$ itself is included  in this equation, but miracles of this type are not excluded!
%and choose units for which $8\pi G=1$, then the TMG equation in the form (\ref{TMG2})  is modified to
%\begin{equation}\label{MMG1}
%C_{\mu\nu} = -(\sigma\mu) S_{\mu\nu} - \left(\mu\Lambda - \frac{1}{4} (\sigma\mu) R\right) g_{\mu\nu} - \frac{\gamma}{\mu} J_{\mu\nu}\, .
%\end{equation}

For the case in hand, if we  take our source tensor to be
\begin{equation}
T_{\mu\nu} = - \frac{\gamma}{\mu^2} J_{\mu\nu}\, , 
\end{equation}
for some dimensionless constant $\gamma$, then the sourced equation implies that the Cotton tensor is now a linear combination of the Schouten tensor, metric tensor and the $J$-tensor. Using this expression, we find that
\begin{equation}
\sqrt{-\det g}\, D_\mu J^{\mu\nu} = -\frac{\gamma}{\mu}\varepsilon^{\nu\rho\sigma} S_\rho{}^\tau J_{\sigma\tau} \equiv 0 \,.
\end{equation}
The final identity is due to the specific  form of $J$ and the symmetry of the Schouten tensor, so the $J$-tensor is {\it still} conserved!  We have now verified the consistency of the ``minimal massive gravity'' equation
\begin{equation}\label{MMG2}
\Lambda g_{\mu\nu} + \sigma G_{\mu\nu} + \frac{1}{\mu} C_{\mu\nu} + \frac{\gamma}{\mu^2} J_{\mu\nu} =0\, , 
\end{equation}
which is still  a third-order  equation because the $J$-tensor is only second order.  This equation provides an explicit and simple example of the third way. It demonstrates that it may be possible to include,  in a gravitational field equation, tensors constructed entirely from the metric that do {\it not} satisfy a Bianchi identity.

In such cases there will be no ``geometrical'' action; i.e.~one constructed entirely from the metric. Nevertheless, at least for our MMG example, there {\it is} an action involving auxiliary fields \cite{Bergshoeff:2014pca}. We will not present it here, although it is fairly simple; instead we address the puzzle of how an action with auxiliary fields is possible when their elimination appears to give us an action for the metric alone, in contradiction to our claim that no such action exists. 
%The answer leads us directly to another manifestation of the third way.

A set of  fields is ``auxiliary'' if the field equations allow us to solve for them  in terms of the other fields. The MMG action is such that the field equations determine the auxiliary fields in terms of the metric and its curvature tensor,  such that the remaining equation for the metric alone is precisely  (\ref{MMG2}). However,  back-substitution into the {\it action}  is legitimate only if the equations used to solve for the  auxiliary fields are equivalent, jointly, to the equations obtained by variation with respect to them. In the MMG case,  the equation obtained by variation of the metric is needed to solve for the  auxiliary fields, and this makes back-substitution into the action illegitimate. This will be the case for any action that yields   ``third-way consistent'' field equations.

To summarise: despite the fact that its left-hand side does {\it not} satisfy a Bianchi identity,  the MMG equation (\ref{MMG2}) {\it is} derivable from an action.  This could have been just a curiosity but it turns out to be essential to the resolution of the ``bulk-boundary clash'' of massive 3D gravity.  Linearization shows that MMG propagates a single massive spin-2 mode and a full Hamiltonian analysis shows that there are no other bulk degrees of freedom,  just like TMG. However, the AdS boundary properties of MMG differ from those of TMG: provided $\gamma<0$ one can choose parameters to arrange for both central charges of the asymptotic conformal symmetry algebra to be positive \cite{Bergshoeff:2014pca}.  Allowing for equivalences,  there is  a single connected region in the MMG parameter space for which all unitary conditions accessible to a semi-classical analysis are satisfied \cite{Arvanitakis:2014xna}.

Third-way consistency has another surprising consequence, which we again illustrate using MMG.  Let us add a source tensor ${\cal T}$ to the right-hand side of the MMG equation:
\begin{equation}
\Lambda g_{\mu\nu} + \sigma G_{\mu\nu} + \frac{1}{\mu} C_{\mu\nu} + \frac{\gamma}{\mu^2} J_{\mu\nu} =  \mathcal{T}_{\mu\nu} \,.
\end{equation}
For  TMG, we may choose ${\cal T}$ to be a matter stress tensor,  but this choice is not possible for non-zero $\gamma$; i.e.~for  MMG. In fact, consistency requires that
\begin{equation}\label{identity}
D_\mu {\cal T}^{\mu\nu} = \frac{\gamma}{\mu\sqrt{-\det g}}\,  \varepsilon^{\nu\rho\sigma} S_\rho{}^\tau {\cal T}_{\sigma\tau}\,  .
\end{equation}
Given a matter stress tensor $T$ one can find a corresponding source tensor ${\cal T}$ that solves this equation \cite{Arvanitakis:2014yja}, but it is quadratic  in $T$!
A corollary is that some solutions of MMG can exhibit qualitatively different behaviour when compared to analogous solutions of TMG because the
tensor ${\cal T}$ need not satisfy energy conditions imposed on  $T$. In particular, big-bang singularities of  TMG cosmological solutions  with an ideal fluid source are absent  in MMG \cite{Arvanitakis:2014yja}.

To conclude, we have illustrated the possibility of a ``third way'' to construct consistent gravitational field equations using the example of ``minimal massive gravity'', and we
have discussed its physical relevance in this context.  All known examples are in 3D, and there is an analog for 3D Yang-Mills theory \cite{Arvanitakis:2015oga},  but the basic idea is dimension independent. It may be  that there exist higher-dimensional gravitational models with field equations that are similarly third-way consistent. In view of this, it is perhaps worth recalling that one of the assumptions of Lovelock's generalisations  of the second-order Einstein field equations  to spacetimes of arbitrary dimension is that all terms satisfy a Bianchi identity \cite{Lovelock:1971yv}. The possibility of a ``third way'' in higher dimensions calls into question this assumption.



\section*{Acknowledgements}

We would like to thank Alex Arvanitakis and Olaf Hohm for their collaboration on work leading to this essay. AJR is supported by a grant from the London Mathematical Society, and he would also like to thank the University of Groningen for hospitality during the writing of this essay.

\providecommand{\href}[2]{#2}\begingroup\raggedright\endgroup

\end{document}
